Author's response to reviews

Title: A systematic literature review of pediculosis due to head lice in Pacific Island Countries and Territories: What country specific research on head lice is needed?

Authors:

Rick Speare (rickspeare@gmail.com)
Humpress Harrington (humpress.harrington@gmail.com)
Deon Canyon (dcanyon@hawaii.edu)
Peter D Massey (Peter.Massey@hnehealth.nsw.gov.au)

Version: 4  Date: 8 June 2014

Author's response to reviews:

Comments in attached Word document.